Notes of the RICH1 Phone meeting 28/09/2007
People Present:
CERN:  Fabio
IC:         Bill, Dave, Richard (from Home), Trevor
Oxford:  Brian
Bristol:   
Apologies for absence:
Neville 
Discussed Agenda Topics
1. Gas Enclosure

1.1 Status

Bill reported on the closure of the GE in preparation for the magnet test. All went smoothly and the leak test (mainly to ensure that the diaphragm was still intact after the beam-pipe bake-out) was satisfactory. The new door bolt-hole plugs appear to work well.

1.2 Temperature Sensors

RWP has rewired the sensors on to a 37 pin D-type connector which will require a feed-through in the patch panel.

1.1 Gas Lines

Christophe and Dave have agreed to install one fixed pipe from the distribution rack to the C-side under RICH1 and one pipe to the top on the A-side. The installation cost for the pipes will be 8KCHF. There is also the possibility to include a stainless steel (and light tight) safety bubbler for an additional 4KCHF. This money is not in the UK RICH1 budget. In addition four flexible pipes will need to be ordered with the expectation that delivery could take several weeks.
2. Preparation for the magnet tests.
The magnetic shielding doors were replaced very rapidly by B.Chadaj with the DBS team and they also did the cleanup of material from underneath RICH1.

The movement of RICH1 during the magnet test will be measured by the surveyors on the A-side and the upper and lower shields will be monitored on the C-side using a theodolyte fixed to the cavern walls and rulers mounted on the iron.

3. Mirrors

Following the last measurement it appears that the only parameter not within satisfactory limits is the rotation of the spherical mirrors about the Y axis where no adjustment was foreseen. No further work is currently planned. 
4. HPD assembly
4.1 Column dressing

RWP has completed all the ladder assemblies apart from the two spares. Ken has made an audit of all the available parts and because of the shortage of spare HPDs and Level0 boards it has been decided that none will be fitted to the spare ladders. The LV and HV cards have been installed but not the cables or Level 0 cards.

4.2 Upper HPD rail system
Brian reported good progress on the modifications being made in the Oxford workshop but an estimated delivery date was not available (following the meeting 5 October was confirmed). The new support cable is ordered but not yet delivered. There was no information on the situation of the new support beam at NAB.

4.3 Lower HPD rail system
The drawings for two of the frames have been delivered to the workshop where they are waiting for a welder to become available. The material is available. Brian hopes to deliver the next drawing by Tuesday when he will then require the rail coordinates from Trevor in order to make further progress. These have been promised by 5 October.
4.4 HPD cabling
The data fibres have still to be installed and a schedule is required for connector fitting by Maria & Vera. 
4.5 Time alignment laser

The plan is to have two fibres in each HPD module pointing at the opposite side. The fibres will not be column spares but part of a new bundle which therefore requires breakouts to be bought and allocated space in the box.
4.6 Light leak detector
The PM tubes have not been delivered yet.

DMW explained that the system in RICH2 had some difficulties to be resolved because the light emitted by the timing laser was sufficient to trigger the light leak system and turn off the high voltage. This may be true even of the expected amount of Cherenkov light, so careful design of the dynamic range of the leak detection system is required.

4.7 Temperature monitors

RWP explained that these will be glued inside the hood and taken to planned sockets on the patch panels. They do not have to be removed for column extraction.  
4.8 Photon funnel

It should now be possible to proceed with the final design when engineering effort becomes available. The RAL engineers would have preferred to have performed a test with glass before proceeding with the quartz but they have now accepted that the situation is not the same as RICH2. Derek needs four weeks to mount the quartz and that fact, combined with the expected delivery time, means that the order for the photon funnels must be placed this year.  
4.9 Mechanics

The main parts of the HPD chariot have been sent off to be anodized and other pieces are being machined in the workshop. It appears almost certain that the mid-October slot for an installation test with the complete box will be missed.

5.0 LAMS

RWP is making good progress with the design of the mounting brackets.

6.0 MDMS

Dave has received a summary of recent progress from Ray. He has been distracted by the Vertex 2007 Conference but estimated to deliver a cross-section drawing of the mechanics within one week. He is still working on the design of the electronic system.

Email from Ray:

As discussed at our last meeting, I have been revisiting the mechanical design, specifically the support and cable feedthroughs,  in order that Trevor may finalize the remaining details of the photon funnel design for manufacture.  A proposal will be circulated shortly.  The design for the light source electronics and optical components is proceeding.  Estimate one week for schematic.

9. Next meeting
Friday 12th October 2007 at 10:00(GMT) 11:00(CET). 
